Author's response to reviews

Title: Herpes encephalitis presenting as mild aphasia

Authors:

Omar A. Khan (omar.khan@vtmednet.org)
Allan Ramsay (allan.ramsay@vtmednet.org)

Version: 4 Date: 14 March 2006

Author's response to reviews:

Dear Editor:

Thank you for your constructive feedback.

>>Please structure your abstract, if possible add a small background section to it.

The abstract has been re-structured per BMC guidelines.

>>Manuscript: Please add a conclusions section after the case presentation, this can be quite small.

This has been done as well.

Please see separate email which includes patient consent for the Case Report.

Sincerely,

Omar Khan
Allan Ramsay